Author's response to reviews

Title: Redo-redo aortic root replacement with a mechanical valved conduit in a patient with von Willebrand's disease: Case report

Authors:

Kasra Shaikhrezai (kasrash@gmail.com)
Usman Bashir (usmanbashir@nhs.net)
Sheena Millar (sheena.millar@luht.scot.nhs.uk)
Julia Anderson (Julia.anderson@luht.scot.nhs.uk)
Edward T Brackenbury (edward.brackenbury@luht.scot.nhs.uk)

Version: 2 Date: 12 August 2010

Author's response to reviews:

Dear Editor

We have implemented the requested changes and as there were no available high quality images we have excluded the images files from the manuscript without affecting the content.

Regards
Kasra shaikhrezai et al